\documentclass[12pt]{article}
\usepackage{e-jc}

%\usepackage{geometry}                % See geometry.pdf to learn the layout options. There are lots.
%\geometry{letterpaper}                   % ... or a4paper or a5paper or ... 
%\geometry{landscape}                % Activate for for rotated page geometry
%\usepackage[parfill]{parskip}    % Activate to begin paragraphs with an empty line rather than an indent
\usepackage{graphicx}

\usepackage{amsthm,amsmath,amssymb}

\usepackage{tikz}
\usepackage{color}
\usepackage{mathrsfs}
\usepackage{longtable}

\usepackage{float}

%%%%%Japanese stuff%%%%%%%%%%%%%%%%%%
%\usepackage{fontspec}
%\setmainfont{CMU Serif}
%\usepackage{xeCJK}
%\xeCJKsetup{CJKmath=true}
%\setCJKmainfont{IPAexMincho} % or some other Japanese font
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\theoremstyle{plain}
\newtheorem{theorem}{Theorem}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{fact}[theorem]{Fact}
\newtheorem{observation}[theorem]{Observation}
\newtheorem{claim}[theorem]{Claim}

\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{conjecture}[theorem]{Conjecture}
\newtheorem{open}[theorem]{Open Problem}
\newtheorem{problem}[theorem]{Problem}
\newtheorem{question}[theorem]{Question}

\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\newtheorem{note}[theorem]{Note}

\newcommand{\I}{\mathcal{I}}

\newcommand{\A}{\mathcal{A}}
\newcommand{\B}{\mathcal{B}}
\newcommand{\M}{\mathcal{M}}
\newcommand{\N}{N}
\newcommand{\NN}{\mathbb{N}}
\newcommand{\ZZ}{\mathbb{Z}}
\renewcommand{\L}{\mathcal{L}}
\newcommand{\MLA}{\M_{\A}}
\newcommand{\C}{\mathcal{C}}
\newcommand{\CC}{\mathbb{C}}
\newcommand{\Z}{\mathbb{Z}}
\newcommand{\F}{\mathbb{F}}
\newcommand{\II}{\mathbb{I}}
\newcommand{\V}{\mathbb{V}}
\newcommand{\G}{\mathbb{G}}
\newcommand{\R}{\ensuremath \mathbb{R}}
\renewcommand{\P}{\ensuremath \mathbb{P}}
\renewcommand{\S}{\ensuremath \mathcal{S}}
%\renewcommand{\}{\partial}
\newcommand{\Sec}{\ensuremath{ \text{Sec}}}
\newcommand{\dedication}[1]{\begin{center} {\em #1} \end{center}}
\newcommand{\X}{\ensuremath{ \mathbb{X}}}
\renewcommand{\th}{$^\text{th}$ }
\newcommand{\NO}{NO}
\newcommand{\NU}{NU}
\newcommand{\codim}{\text{codim}}
\newcommand{\h}[4]{\ensuremath{h_p^{#1}\sigma_{#2} \sigma_{#3} h^{#4}}}

\newcommand{\OT}{\mathrm{OT}}
\newcommand{\CP}{\mathbb{CP}}


\renewcommand{\H}{\mathcal{H}}

\renewcommand{\d}{\mathrm{d}}
\newcommand{\wt}{\mathrm{wt}}
\renewcommand{\codim}{\mathrm{codim}}
\renewcommand{\max}{\mathrm{max}}
\renewcommand{\min}{\mathrm{min}}

\newcommand{\rk}{\mathrm{rk}}
\newcommand{\calP}{\mathcal{P}}


\newcommand{\im}{\mathrm{im}}
\newcommand{\bs}{\backslash}
\newcommand{\Fl}{\mathcal{F}l}


%\newcommand{\mod}{\ (\mathrm{mod}}

\newcommand{\ga}{\ensuremath{\alpha}}
\newcommand{\gb}{\ensuremath{\beta}}
\newcommand{\gc}{\ensuremath{\chi}}
\newcommand{\gd}{\ensuremath{\delta}}
\renewcommand{\ge}{\ensuremath{\varepsilon}}
\newcommand{\gfu}{\ensuremath{\phi}}
\newcommand{\gf}{\ensuremath{\varphi}}
\renewcommand{\gg}{\ensuremath{\gamma}}
\newcommand{\gh}{\ensuremath{\eta}}
\newcommand{\gi}{\ensuremath{\iota}}
\newcommand{\gk}{\ensuremath{\kappa}}
\newcommand{\gl}{\ensuremath{\lambda}}
\newcommand{\gm}{\ensuremath{\mu}}
\newcommand{\gn}{\ensuremath{\nu}}
\newcommand{\go}{\ensuremath{\omega}}
\newcommand{\gq}{\ensuremath{\theta}}
\newcommand{\gr}{\ensuremath{\rho}}
\newcommand{\gs}{\ensuremath{\sigma}}
\newcommand{\gt}{\ensuremath{\tau}}
\newcommand{\gu}{\ensuremath{\upsilon}}
\newcommand{\gx}{\ensuremath{\xi}}
\newcommand{\gy}{\ensuremath{\psi}}
\newcommand{\gz}{\ensuremath{\zeta}}



%\DeclareGraphicsRule{.tif}{png}{.png}{`convert #1 `dirname #1`/`basename #1 .tif`.png}



\title{A Flag Whitney number formula for matroid Kazhdan-Lusztig polynomials}


\author{Max Wakefield
\thanks{Supported by the Simons Foundation, the Office of Naval Research, and the Japanese Society for the Promotion of Science.}\\
\small Department of Mathematics\\[-0.8ex]
\small US Naval Academy\\[-0.8ex]
\small Annapolis, USA\\
\small wakefiel@usna.edu}

\date{\dateline{May 7, 2016}{November 20, 2017}\\
\small Mathematics Subject Classifications: 05B35, 06A07, 11B75}


\begin{document}
\begin{thebibliography}{10}

\bibitem{AHK17}
Karim Adiprasito, June Huh, and Eric Katz.
\newblock Hodge theory for combinatorial geometries.
\newblock preprint, arXiv:1511.02888.

\bibitem{A87}
Martin Aigner.
\newblock Whitney numbers.
\newblock In {\em Combinatorial geometries}, volume~29 of {\em Encyclopedia
  Math. Appl.}, pages 139--160. Cambridge Univ. Press, Cambridge, 1987.

\bibitem{bren03}
Francesco Brenti.
\newblock {$P$}-kernels, {IC} bases and {K}azhdan-{L}usztig polynomials.
\newblock {\em J. Algebra}, 259(2):613--627, 2003.

\bibitem{sage}
The~Sage Developers.
\newblock {\em {S}age {M}athematics {S}oftware ({V}ersion 6.6)}, 2016.
\newblock {\tt http://www.sagemath.org}.

\bibitem{DW75}
Thomas~A. Dowling and Richard~M. Wilson.
\newblock Whitney number inequalities for geometric lattices.
\newblock {\em Proc. Amer. Math. Soc.}, 47:504--512, 1975.

\bibitem{GPY}
Katie Gedeon, Nicholas Proudfoot, and Ben Young.
\newblock {The equivariant Kazhdan-Lusztig polynomial of a matroid}.
\newblock preprint.

\bibitem{GZ83}
Curtis Greene and Thomas Zaslavsky.
\newblock On the interpretation of {W}hitney numbers through arrangements of
  hyperplanes, zonotopes, non-{R}adon partitions, and orientations of graphs.
\newblock {\em Trans. Amer. Math. Soc.}, 280(1):97--126, 1983.

\bibitem{hall}
P.~Hall.
\newblock A {C}ontribution to the {T}heory of {G}roups of {P}rime-{P}ower
  {O}rder.
\newblock {\em Proc. London Math. Soc.}, S2-36(1):29.

\bibitem{HW17}
June Huh and Botong Wang.
\newblock Enumeration of points, lines, planes, etc.
\newblock preprint, arXiv:1609.05484v3.

\bibitem{Karu06}
Kalle Karu.
\newblock The {$cd$}-index of fans and posets.
\newblock {\em Compos. Math.}, 142(3):701--718, 2006.

\bibitem{KL79}
David Kazhdan and George Lusztig.
\newblock Representations of {C}oxeter groups and {H}ecke algebras.
\newblock {\em Invent. Math.}, 53(2):165--184, 1979.

\bibitem{Kung86}
Joseph P.~S. Kung.
\newblock Radon transforms in combinatorics and lattice theory.
\newblock In {\em Combinatorics and ordered sets ({A}rcata, {C}alif., 1985)},
  volume~57 of {\em Contemp. Math.}, pages 33--74. Amer. Math. Soc.,
  Providence, RI, 1986.

\bibitem{EPW15}
Nicholas Proudfoot, Max Wakefield, and Ben Young.
\newblock Intersection cohomology of the symmetric reciprocal plane.
\newblock {\em J. Algebraic Combin.}, 43(1):129--138, 2016.

\bibitem{PWY16}
Nicholas Proudfoot, Max Wakefield, and Ben Young.
\newblock Intersection cohomology of the symmetric reciprocal plane.
\newblock {\em J. Algebraic Combin.}, 43(1):129--138, 2016.

\bibitem{PXY}
Nicholas Proudfoot, Yuan Xu, and Ben Young.
\newblock {The Z-polynomial of a matroid}.
\newblock preprint.

\bibitem{rota64}
Gian-Carlo Rota.
\newblock On the foundations of combinatorial theory. {I}. {T}heory of
  {M}\"obius functions.
\newblock {\em Z. Wahrscheinlichkeitstheorie und Verw. Gebiete}, 2:340--368
  (1964), 1964.

\bibitem{Seym82}
P.~D. Seymour.
\newblock On the points-lines-planes conjecture.
\newblock {\em J. Combin. Theory Ser. B}, 33(1):17--26, 1982.

\bibitem{stan72B}
R.~P. Stanley.
\newblock Supersolvable lattices.
\newblock {\em Algebra Universalis}, 2:197--217, 1972.

\bibitem{stan72A}
Richard~P. Stanley.
\newblock {\em Ordered structures and partitions}.
\newblock American Mathematical Society, Providence, R.I., 1972.
\newblock Memoirs of the American Mathematical Society, No. 119.

\bibitem{stan74}
Richard~P. Stanley.
\newblock Finite lattices and {J}ordan-{H}\"older sets.
\newblock {\em Algebra Universalis}, 4:361--371, 1974.

\bibitem{stan79}
Richard~P. Stanley.
\newblock Balanced {C}ohen-{M}acaulay complexes.
\newblock {\em Trans. Amer. Math. Soc.}, 249(1):139--157, 1979.

\bibitem{stan92}
Richard~P. Stanley.
\newblock Subdivisions and local {$h$}-vectors.
\newblock {\em J. Amer. Math. Soc.}, 5(4):805--851, 1992.

\bibitem{Stan94}
Richard~P. Stanley.
\newblock Flag {$f$}-vectors and the {$cd$}-index.
\newblock {\em Math. Z.}, 216(3):483--499, 1994.

\bibitem{W76}
D.~J.~A. Welsh.
\newblock {\em Matroid theory}.
\newblock Academic Press [Harcourt Brace Jovanovich, Publishers], London-New
  York, 1976.
\newblock L. M. S. Monographs, No. 8.

\end{thebibliography}


\maketitle

\begin{abstract} 

For a representation of a matroid the combinatorially defined Kazhdan-Lusztig polynomial computes the intersection cohomology of the associated reciprocal plane. However, these polynomials are difficult to compute and there are numerous open conjectures about their structure. For example, it is unknown whether or not the coefficients are non-negative for non-representable matroids. The main result in this note is a combinatorial formula for the coefficients of these matroid Kazhdan-Lusztig polynomials in terms of flag Whitney numbers. This formula gives insight into some vanishing behavior of the matroid Kazhdan-Lusztig polynomials.

\end{abstract}

\section{Introduction}

Flag Whitney numbers of the second kind are key players in the study of the cd-index and are usually grouped together into one function called the flag f-vector (see \cite{Stan94}). They count chains in a ranked partially ordered set with prescribed ranks. There are many open conjectures about classical Whitney numbers of the first and second kind (for example \cite{Seym82} and \cite{DW75}). Some parts of these conjectures have just been recently resolved. In particular, Adiprasito, Huh, and Katz in \cite{AHK17} proved the log-concavity conjecture for arbitrary matroids (see 15.3 exercise 5 in \cite{W76}) and Huh and Wang in \cite{HW17} proved the top heaviness conjecture (see \cite{Kung86}) for representable matroids. There are also some classical results, like the generalized ``Hyperplane Theorem'' (see \cite{DW75}). However, there is still much unknown about Whitney numbers, like the points-lines-planes conjecture (see \cite{Seym82}). 

In this note, flag Whitney numbers are used to understand the Kazhdan-Lusztig polynomial of a matroid or more generally of a finite ranked poset. The Kazhdan-Lusztig polynomial of a matroid, first studied in \cite{EPW15}, is a single variable polynomial defined by a recursion which mimics the classical Kazhdan-Lusztig polynomials (see \cite{KL79}). In the combinatorial recursion the characteristic polynomial takes the place of the classical ``$R$-polynomials''. It turns out that these matroid Kazhdan-Lusztig polynomials sit in the general framework of Kazhdan-Lusztig-Stanley polynomials as developed by Stanley in \cite{stan92} and refined by Brenti in \cite{bren03}. If the matroid is realizable over $\CC$ then the matroid Kazhdan-Lusztig polynomial is the intersection cohomology Poincar\'e polynomial of the associated reciprocal plane (again see \cite{EPW15}). These polynomials are notoriously difficult to compute in general. For example, the main focus in \cite{PWY16} is the intersection cohomology for uniform matroids of rank $n-1$ on $n$ elements. In \cite{GPY} an equivariant matroid Kazhdan-Lusztig polynomial is defined and impressively used to find a formula for the ordinary matroid Kazhdan-Lusztig polynomial for all uniform matroids. 

Our main result, Theorem \ref{KLpolymain}, is a formula for the coefficients of the posets Kazhdan-Lusztig polynomial in terms of sums of ``top-heavy pairs'' of flag Whitney numbers of the second kind (see Section \ref{whitneysec}). The crux of this formula is developing the index set to which we sum over. At this point this index set is written down recursively, however probably with a little more work one can write it down in a closed form. In \cite{EPW15} the degree 1 and 2 coefficients of the Kazhdan-Lusztig polynomial of a matroid are written in terms of doubly-indexed Whitney numbers of the second kind (see \cite{GZ83}). Theorem \ref{KLpolymain} extends those formulas to arbitrary coefficients. This ``top-heavy'' formula gives a clear picture of why these polynomial's positive degree coefficients vanish when the poset is modular. Recently, Proudfoot, Xu, and Young in \cite{PXY} have found a formula similar to Theorem \ref{KLpolymain} using a completely different technique. Their result is not decomposed in terms of top-heavy pairs, but is stated much more concretely and in closed form. 

The paper is organized as follows. In Section \ref{whitneysec} some basic facts concerning flag Whitney numbers are collected. The central focus there is the celebrated Theorem of Phillip Hall \cite{hall} on the Euler characteristic of a finite lattice. Then in Section \ref{KLpolysec} we review the definition of matroid Kazhdan-Lusztig polynomials and some basic combinatorial results. In section \ref{index-sec} we create the index set used to state Theorem \ref{KLpolymain}. Finally in Section \ref{thm} we state the main theorem and note a consequence and write down the formulas for coefficients up to degree 5.

\section{Whitney numbers}\label{whitneysec}

In this section we record combinatorial results on Whitney numbers for use on the matroid Kazhdan-Lusztig polynomials. Let $\calP$ be a ranked locally finite poset. For the remainder of the paper we use $I=\{i_1,\dots ,i_k\}$ (or other $k$-tuples) to denote an ordered (by usual $\leq$) $k$-tuple such that for all $j$, $i_j\in \{0,1,2,\dots, \rk \calP\}$. The set of partial flags in $\calP$ indexed by $I$ is $$\calP_I=\{(X_1,\dots,X_k)\in \calP^k\ |\ \forall \ 1\leq j\leq k, \rk (X_j)=i_j\}.$$

\begin{definition}

The \emph{partial flag Whitney numbers of the second kind indexed by $I$} are $$W_I(\calP)=|\calP_I|.$$ When the context of the poset is clear we will just write $W_I$ instead of $W_I(\calP)$.
\end{definition}

The classical Whitney numbers of the second kind are those with $k=1$ and in this case we just write $W_i$. A good reference for classical Whitney numbers is \cite{A87} and these were generalized to 2 subscripts in \cite{GZ83}. These flag Whitney numbers of the second kind have the property that some indices are ``trivial''. For example, if $\calP$ is a lattice then $W_{0,i}=W_i$.

\begin{remark}

What we call flag Whitney numbers here were studied by Richard Stanley in multiple papers \cite{stan72A}, \cite{stan72B}, \cite{stan74}, and  \cite{stan79} where they were denoted by $\alpha(\calP ,I)$, but they were only considered for strictly increasing flags. Here we study flags where some coordinates may be equal. 

\end{remark}

\begin{example}\label{W(B_n)}

Let $\B_n$ be the rank $n$ Boolean lattice. For just this example we let $\B_n(j)=(\B_n)_j$ the rank $j$ component of $\B_n$. If $I=(i_1,\dots ,i_k)$ then $$W_I(\B_n)=\binom{n}{i_1,i_2-i_1,i_3-i_2,\dots ,i_k-i_{k-1},n-i_k} .$$

\begin{proof}

We start at the bottom of the chain $i_1\leq \cdots \leq i_k$. There are exactly $\binom{n}{ i_1}$ elements of rank $i_1$ in $\B_n$ (i.e. $|\B_n(j)|=\binom{n}{j}$). Then for any $X\in \B_n(i_1)$ the restriction to $X$ is $\B_n^X\cong \B_{n-i_1}$ and the elements above $X$ of rank $i_2$ in $\B_n$ are now of rank $i_2-i_1$ in $\B_n^X$. So, for every $X\in \B_n(i_1)$ the number of elements above it is $\binom{n-i_1}{i_2-i_1}$. In general for every $Y\in \B_N(i_j)$ there are $\binom{n-i_j}{i_{j+1}-i_j}$ above it in $\B_n(i_{j+1})$. Hence $$W_I(\B_n)=\binom{n}{i_1} \binom{n-i_2}{i_2-i_1}\cdots \binom{n-i_{k-1}}{ i_k-i_{k-1}}=\binom{n}{ i_1,i_2-i_1,i_3-i_2,\dots ,i_k-i_{k-1},n-i_k}.$$\end{proof}

\end{example}

Now we work towards a formula that relates the multi-indexed Whitney numbers of second kind to that of the first. The Whitney numbers of the first kind are built using the M\"obius function on the poset. The M\"obius function on a poset $\calP$ is $\mu:\calP\times \calP\to \ZZ$ defined recursively by $\mu(X,X)=1$ and $$\mu (X,Y)=-\sum\limits_{X\leq Z< Y}\mu (X,Z).$$ For a locally finite ranked poset with minimal element $\hat{0}$ the classical Whitney numbers of the first kind are the numbers $$w_i=\sum\limits_{X\in \calP_i} \mu (\hat{0}, X).$$ Let $\calP$ be a locally finite ranked poset with smallest element $\hat{0}$. All the elements of $\calP$ of rank $k$ we denote by $\calP_k:=\{X\in \calP\ |\  \rk X=k\}$ and for $I=\{i_1,\dots ,i_s\}$ we set $\calP(I)=\{\vec{X}=(X_1,\dots ,X_s)\ | \ \forall 1\leq i\leq s, \ X_i\in \calP(i)\}$. Also, we call $\calP_X=\{Y\in \calP\ |\ Y\leq X\}$ the localization of $\calP$ at $X$ and $\calP^X=\{Y\in \calP \ |\  Y\geq X\}$ the restriction of $\calP$ at $X$. Using these new posets we record a few basic lemmas which are foundational for computing various Whitney numbers.

\begin{lemma}\label{add1}

If $I\subseteq \{1,\dots ,n-1\}$ then $$\sum\limits_{X\in\calP_n}W_{I}(\calP_X)=W_{I\cup \{n\}}(\calP).$$

\end{lemma}

\begin{proof}

By definition $$W_I(\calP_X)=\sum\limits_{\substack{\vec{X}\in\calP (I)\\ \vec{X}\leq X}}\zeta (\vec{X}).$$ Thus \begin{equation}\label{final1}\sum\limits_{X\in\calP_n}W_{I}(\calP_X)=\sum\limits_{X\in\calP_n}\sum\limits_{\substack{\vec{X}\in\calP (I)\\ \vec{X}\leq X}}\zeta (\vec{X})=\sum\limits_{X\in\calP_n}\sum\limits_{\vec{X}\in\calP (I)}\zeta (\vec{X},X). \end{equation} Since the right hand side of \ref{final1} is exactly $W_{I\cup \{n\}}$ we are done.\end{proof}

We add the ``dual'' of Lemma \ref{add1} for later use whose proof is very similar.

\begin{lemma}\label{add2}

Let $r=\rk (\calP)$ and $I\subseteq \{1,\dots ,t-1\}$. For $I=\{i_1,\dots ,i_k\}$ set $I[t]=\{i_1+t,i_2+t,\dots ,i_k+t\}$ and assume that $i+t\leq r$ for all $i\in I$. With this notation we have $$\sum\limits_{X\in\calP_{t}}W_{I}(\calP^X)=W_{\{t\}\cup I[t]}(\calP).$$

\end{lemma}

\begin{proof}

In this case the indices $I$ must be shifted to be accounted for in $\calP$ because $\calP^X$ is all elements above $X$. So, $$\sum\limits_{X\in \calP_t}W_I(\calP^X)=\sum\limits_{X\in \calP_t}\sum\limits_{\vec{X}\in \calP^X(I)}\zeta(\vec{X})$$ $$=\sum\limits_{X\in \calP_t}\sum\limits_{\substack{ \vec{X}\in \calP (I+t) \\ X\leq \vec{X}}}\zeta(\vec{X})=\sum\limits_{X\in \calP_t}\sum\limits_{\vec{X}\in \calP(I+t)}\zeta (X,\vec{X})=\sum\limits_{\vec{Y}\in (\{t\}\cup (I+t))}\zeta (\vec{Y})$$ which is exactly $W_{\{t\}\cup (I+t)}$.\end{proof}

We add another lemma for use on understanding the Kazhdan-Lusztig polynomial, which is really a combination of Lemma \ref{add1} and Lemma \ref{add2}.

\begin{lemma}\label{whitprod}

Let $r=\rk(\calP)$, $k\in [r]$ and $I,J\subseteq [r]$ such that for all $i\in I$, $i\leq k$ and for all $j\in J$, $j+k\leq r$. In this case we can define $J[k]=\{j+k\ |\ j\in J\}$. Then $$\sum\limits_{F\in L_k}W_I(\calP_F)W_J(\calP^F)=W_{I\cup \{k\} \cup J[k]}(\calP).$$

\end{lemma}

\begin{proof}

Let $I=\{i_1,\dots ,i_s\}$ and $J=\{j_1,\dots ,j_t\}$. Look at the sum 

$$\sum\limits_{F\in L_k}W_I(\calP_F)W_J(\calP^F)=\sum\limits_{F\in \calP_k}\Bigg(\sum\limits_{\mathscr{X}}1\Bigg)\Bigg(\sum\limits_{\mathscr{Y}}1\Bigg)$$ where the summation condition $\mathscr{X}$ correspond to $X_u\in \calP_{i_u}$ for $1\leq u\leq s$ and $X_{i_1}\leq \cdots X_{i_s}\leq F$ and the summation condition $\mathscr{Y}$ corresponds to $Y_v\in \calP_{j_v}$ for $1\leq v\leq t$ and $F\leq Y_{j_1}\leq \cdots \leq Y_{j_t}$. Then we switch the sums and we have the result.\end{proof}


Now we present the main lemma of this section. This is proved on page 154 of \cite{stan79} and can follow by using the previous Lemmas. It can also be thought of as a consequence of Phillip Hall's Theorem (see \cite{hall} and \cite{rota64}, Proposition 6).

\begin{lemma}\label{interpolation}

If $\calP$ is a locally finite, ranked poset  with a minimal element and $1\leq n\leq \rk (\calP)$ then $$w_{0,n}=\sum\limits_{I\subseteq \{1,\dots ,n-1\}}(-1)^{|I|+1}W_{I\cup \{n\}} .$$

\end{lemma}

\section{The Kazhdan-Lusztig polynomial of a matroid}\label{KLpolysec}

The aim of this section is to develop formulas for certain coefficients of the Kazhdan-Lusztig polynomial of a matroid. This result gives some hint that these polynomials may be more tractable to understand than the classical Kazhdan-Lusztig polynomials. These matroid Kazhdan-Lusztig polynomials were originally defined for matroids or equivalently for geometric lattices. However they can be defined for any finite ranked poset with bottom and top elements. To do this we need a little notation. Let $\calP$  be a finite ranked poset with a top element $\hat{0}$ and bottom element $\hat{1}$. For $F\in \calP$ the restriction of $\calP$ to $F$ is $$\calP^F=\{E\in \calP\ |\  E\geq F\}$$ and the localization of $\calP$ at $F$ is $$\calP_F=\{E\in P\ |\ E\leq F\} .$$ The last ingredient we need is the infamous characteristic polynomial.

$$\chi (\calP, t)=\sum\limits_{X\in \calP}\mu (\hat{0},X)t^{\rk (X)}.$$ The following definition of matroid Kazhdan-Lusztig polynomials requires a fascinating property of the characteristic polynomial, that it is a $\calP$-kernel in the sense of \cite{bren03} and \cite{stan92}. 

\begin{definition}[\cite{EPW15}, Theorem 2.2]\label{KLdef}

Let $\calP$ be a finite ranked poset with top and bottom. The \emph{Kazhdan-Lusztig polynomial} of $\calP$, $P(\calP,t)$ is the polynomial recursively defined which satisfies

\begin{enumerate}

\item If $\rk(\calP)=0$ then $P(\calP,t)=1$.

\item If $\rk (\calP)>0$ then $\deg (P(\calP,t))<.5\rk(\calP)$.

\item For all $\calP$, \begin{equation*}\label{recursion} t^{\rk(\calP)}P(\calP,t^{-1})=\sum\limits_{F\in \calP}\chi(\calP_F,t)P(\calP^F,t) .\end{equation*} 

\end{enumerate}

\end{definition}

Now we gather some basic results on the first few coefficients from \cite{EPW15}.

\begin{proposition}[\cite{EPW15}, Propositions 2.11, 2.12, and 2.16]\label{0,1,2}

Let $L$ be a geometric lattice with rank $r$. Then 

\begin{enumerate}

\item The constant coefficient of $P(L,t)$ is 1.

\item The linear coefficient of $P(L,t)$ is $W_{r-1}-W_{1}$.

\item The quadratic coefficient of $P(L,t)$ is $$-[W_{1,r -1}-W_{1,2}]+[W_{r-3,r-1}-W_{r-3,r-2}]+[W_{r-2}-W_{2}].$$ 
\end{enumerate}
\end{proposition}

Using Proposition \ref{0,1,2} for any matroid of rank less than 6 we can quickly calculate the Kazhdan-Lusztig polynomials. However, a central aim in this field is to compute the Kazhdan-Lusztig polynomials for certain infinite families of arrangements with higher ranks. For example, a closed formula of the Kazhdan-Lusztig polynomials for the family of braid matroids or reflection types is unknown. Quite surprisingly the only infinite family of non-trivial matroids with a known closed formula are the uniform matroids (see \cite{PWY16} and \cite{GPY}). One aim of this note is to generalize Proposition \ref{0,1,2} in order to compute the Kazhdan-Lusztig polynomials for some families of matroids.

\section{The index set}\label{index-sec}

In this section we develop some notation to state a formula for any coefficient of the Kazhdan-Lusztig polynomial. We are going to compute the degree $k$ term. Developing the index set to sum over is the crux. Throughout we denote $\{1,\dots ,n\}$ by $[n]$. We are going to define the index set, which we will call $S_k$, recursively. Eventually $S_k$ will be a set of subsets of $\ZZ [r]$ where we will view $r$ as a variable. The elements in each subset will either be an integer or of the form $r-n$ for an integer $n$. Then later when we want to use the formula the aim will be to substitute the rank of the matroid or poset for $r$. The base is $S_1=\{\{1\}\}$. For $1\leq t$ put $$A_{t}=\Big\{I\in 2^{[t]}\Big| t\in I \Big\}$$ and for $t\leq 0$ we set $A_t=\emptyset$. So, for example $A_1=\{\{1\}\}$, $A_2=\{\{2\},\{1,2\}\}$, and $A_3=\{\{3\},\{1,3\},\{2,3\},\{1,2,3\}\}$.

Next for $3\leq s\leq 2k-1$ we need a function $f_s:\ZZ [r]\to \ZZ [r]$ defined by $$f_s(p)=\mathrm{eval}(p,s)+r-s$$ where $\mathrm{eval}(p,s)$ means evaluating the polynomial $p$ at $s$. Then for a finite subset $I$ of $\ZZ[r]$ put $F_s(I)=\{f_s(i)\ |\ i\in I\}$. For example, $F_4(\{1,r-2,r-1\})=\{r-3,r-2,r-1\}$. Then for $k>1$ set \begin{equation}\label{mainrecursion}S_k=A_k\cup \bigcup\limits_{s=3}^{2k-1} T_k^s\end{equation} where \begin{equation}\label{unions}T_k^s=\bigcup\limits_{u\leq i<s/2} T_k^s(i)\end{equation} with $u=\mathrm{max}\{1,s-k\}$ and \begin{equation}\label{detailrecursion}T_k^s(i)=\Big\{ \alpha \sqcup \{r-s\} \sqcup F_s(\beta )\Big| \alpha\in A_{k-s+i},\ \gb\in S_i \Big\} .\end{equation} It is important to make sure $S_k$ is well defined. Note that $i<s/2<k$ implies the sets $T_k^s(i)$ are well defined and hence means $S_k$ is well defined.

For example, with $k=2$ we have $S_2=A_2\cup T_2^3$, $u=1$, $T_2^3=T_3^3(1)$, and \begin{equation}\label{k=2recursion}T_2^3(1)=\Big\{ \alpha \sqcup \{r-3\} \sqcup F_3(\beta )\Big| \alpha\in A_{0}=\emptyset,\ \gb\in S_1 \Big\} =\{\{r-3,r-2\}\}.\end{equation} Hence $S_2=\{\{2\},\{1,2\}, \{r-3,r-2\}\}$ and note that these are exactly the index sets of the right hand terms in each bracket of (3) in Proposition \ref{0,1,2}. This is the aim of this index set. It will give the ``bottom terms'' in the ``top heavy pairs'' for our main formula.

Now we need a technical lemma to finish the remainder of the construction. This lemma is a key piece of the entire formula. To prove this lemma we will need a little notation. For $I\in T_k^s$ let $$\max^r(I)=\max\{n\in \ZZ\ |\  r-n\in I\}$$ and similarly $$\min^r(I)=\min\{n\in \ZZ\ |\  r-n\in I\}.$$ 

\begin{lemma}\label{injectiverec}

For each $I=\alpha \sqcup\{r-s\}\sqcup F_s(\beta) \in T_k^s$ there does not exist a different $\alpha'$ and $\beta'$ such that $I=\alpha'\sqcup\{r-s\}\sqcup F_s(\beta')$. 

\end{lemma}

\begin{proof}

First we need a few facts about the elements of $I\in T_k^s$. By induction on $k\geq2$ we show that \begin{equation}\label{eqmaxr}\max\{\max^r(I)\ |\ I\in S_k\}=2k-1.\end{equation} The base case is when $k=2$ and we computed above that $S_2=\{\{2\},\{1,2\}, \{r-3,r-2\}\}$. Hence, $\max\{\max^r(I)\ |\ I\in S_2\}=3=2\cdot 2-1$. Now suppose that $k>2$. Since $\max^r(I)=0$ for $I\in 2^{[k-1]}$ we may assume that $I\in T_k^s$ for some $3\leq s\leq 2k-1$. So, $I=\alpha \sqcup \{r-s\} \sqcup F_s( \beta)$ where $\alpha\in 2^{[k-s+i]}$ and $\beta \in S_i$ for some $\max \{1,k-s\}\leq i\leq s/2$. Since $\alpha$ has no $r$ variables we only need to consider $F_s(\beta)$. By induction since $i<k$ we have $\max^r (\beta)\leq 2i-1<s$. Hence $\max^r(I)=s$ and since the maximum that $s$ can be is $2k-1$ we have finished proving (\ref{eqmaxr}). Notice that within this proof we have also concluded that \begin{equation}\label{maxrplain} \max^r(I)=s\end{equation} for $I\in T_k^s$. If $I\in T_k^s$ is of the form $I=\ga \sqcup \{r-s\}\sqcup F_s(\gb)$, since the elements of $\alpha$ are all integers and $F_{s'}(I)=F_{s'}(\ga)\sqcup \{r-s\}\sqcup F_s(\gb)$, then we can also conclude that \begin{equation}\label{Fmax}\max^r(F_{s'}(I))=\max\{s,s'-\mathrm{min}(\ga)\}.\end{equation}

Now we show directly that for each $I=\alpha \sqcup\{r-s\}\sqcup F_s(\beta) \in T_k^s$ there does not exist another $\alpha'$ and $\beta'$ such that $I=\alpha'\sqcup\{r-s\}\sqcup F_s(\beta')$. The base case is when $k=2$ and in this case there is only $T_2^3=\{\{r-3,r-2\}\}$ which only has one set as seen above. Now suppose $k>2$ and there was such an $\alpha'$ and $\beta'$. Since the $\alpha$ and $\alpha'$ sets only have integer elements (i.e. no $r$ variables) and all other elements contain the variable $r$ then definitely $\alpha=\alpha'$. Also, $\alpha=\alpha'=J\cup \{k-s+i\}$ for some $J\in 2^{[k-s+i-1]}$. This implies that $|\beta|=|\beta'|$ and $\beta,\beta'\in S_i$ for some $\mathrm{max}\{1,s-k\}\leq i<s/2$. Pairing with (\ref{maxrplain}) we have that $F_s(\beta)=F_s(\beta')$ with $\beta,\beta'\in S_i$. If $\beta,\beta'\in A_i$ then clearly $\beta=\beta'$ since $f_s$ is injective when restricted to just integers. 

Suppose $\beta\in A_i$ and $\beta'\in T_{i}^{s'}$ where $3\leq s'\leq 2i-1$. Then $\min^r(F_s(\gb))=s-i$. Also, there exists $\bar{\ga}\in A_{i-s'+i'}$ and $\bar{\gb}\in S_{i'}$ such that  $\gb'=\bar{\ga}\sqcup \{r-s'\}\sqcup F_{s'}(\bar{\gb})$ where $\max\{1,s'-i\}\leq i'\leq s'/2$. Note that the function $F_s$ is the identity on elements that are outputs from another function $F_{s'}$ because elements in the image set of $F_{s'}$ are of the form $r-d$ where $d$ is an integer and the definition of $F_{s}$ is to take these elements plug in to $r-s$ and then subtract off the $s$. Hence $F_s(\gb')=F_s(\bar{\ga})\sqcup \{r-s'\}\sqcup F_{s'}(\bar{\gb})$. Then $\min^r(F_s(\bar{\ga}))=s-(i-s'+i')>s/2+s'/2>i+i'>i'\geq \min^r(F_{s'}(\bar{\gb}))$ by induction. Hence by induction on $i$, with the base $i=1$ clear from the fact that $S_1=\{\{1\}\}$ and so $\gb=\{1\}$, we have that $\min^r(F_{s'}(\gb'))\leq i$. Since $s-i>i$ we have concluded that it is impossible in this case to have $F_s(\gb)=F_{s'}(\gb')$.

In order to treat this next case we need another general inequality. Suppose $\gb\in T_{i}^{s'}$ with $\gb=\ga \sqcup\{r-s'\}\sqcup F_{s'}(\gl)$ where $\ga\in A_{i-s'-i'}$, $\gl\in S_{i'}$, and $\max\{1,s'-i\}\leq i'<s'/2$. Then picking $s$ such that $\max\{1,s-k\}\leq i<s/2$ we will compute $F_s(\gb)$. Note that $\min^r(F_s(\ga))=s-(i-s'+i')=s-i+s_1-i_1>s/2+s'/2>s'$ and $\max^r\{F_s(\gb\backslash \ga)\}=s'$. Hence \begin{equation}\label{frontbig}\min^r(F_s(\ga))>\max^r(F_s(\gb \backslash \ga)) .\end{equation} Combining this with (\ref{Fmax}) we get \begin{equation}\label{Fmaxt}\max^r(F_{s}(\gb))=s-\mathrm{min}(\ga).\end{equation}

Now we can deal with the next case directly. Suppose that $\gb\in T_{i}^{s_1}$ and $\gb' \in T_i^{s_2}$ where $F_s(\gb)=F_s(\gb')$. So, there exists $i_1$ and $i_2$ satisfying $\max\{1,s_1-i\}\leq i_1\leq s_1/2$ and $\max\{1,s_2-i\}\leq i_2\leq s_2/2$ with $\gb=\ga_1\sqcup \{r-s_1\}\sqcup F_{s_1}(\gl_1)$ and $\gb'=\ga_2\sqcup \{r-s_2\}\sqcup F_{s_2}(\gl_2)$ where $\ga_1\in A_{i-s_1+i_1}$, $\ga_2\in A_{i-s_2+i_2}$, $\gl_1\in S_{i_1}$, and $\gl_2\in S_{i_2}$. Now assume that $s_1\leq s_2$. So, by (\ref{frontbig}) we have that  $\ga_1\supseteq \ga_2$. Then we can consider $\gb \backslash \ga_1$ and $\gb\backslash \ga_2$. If $\ga_2\neq \emptyset$ then by a second induction on $|\gb|=|\gb'|$ (with the base case being $|\gb|=1$ is trivial) we are done. If $\ga_2=\emptyset$ then $F_s(\gb')=\gb'=F_s(\ga_1)\sqcup \{r-s_1\} \sqcup F_{s_1}(\gl_1)$. If $|\ga_1|\geq 1$ then $\gb'\backslash \{ r-s_2\} =F_s(F_{s_2}(\gl_2))=F_s(\ga_1\backslash \{\min \{\ga_1\} )\sqcup \{r-s_1\} \sqcup F_{s_1}(\gl_1)$ and again by induction we are done. If $|\ga_1|=\emptyset$ then $s_1=s_2$ and $F_{s_1}(\gl_2)=F_{s_1}(\gl_1)$ and again by induction we are done. \end{proof}

The set $S_k$ will be the index set which we will sum over. But we need to create a ``top heavy'' partner for each index set $I$ to get the full formula. To do this we need a function $d:S_k\to \ZZ[r]$ defined as follows. For $I\in A_k$ define $$d(I)=\left\{ \begin{array}{ll}k& \text{ if } I=\{k\} \\ k-\max \{I\backslash \{k\}\} & \text{ if } I\neq \{k\}\\ \end{array} \right. $$ and for $I\in T_k^s$ define $$d(I)=\min^r(I\backslash \{r-\min^r(I)\})-\min^r(I).$$ For example, $d(\{2\})=2$, $d(\{1,2\})=1$, and $d(\{r-3,r-2\})=\min^r(\{r-3\})-2=1$.
 Finally the ``top heavy'' partner for $I\in S_k$ is $$t(I)=\left\{ \begin{array}{ll} I\backslash \{k\} \cup \{r-d(I)\} & \text{ if } I\in A_k\\ I\backslash \{r-\min^r(I)\} \cup \{r-d(I)\}& \text{ otherwise.}\end{array}\right.$$ For example, $t(\{2\})=\{r-2\}$, $t(\{1,2\})=\{1,r-1\}$, and $t(\{r-3,r-2\})=\{r-3,r-1\}$ and again notice that these are exactly the index sets for the ``top terms'' (left hand) of the ``top heavy'' pairs in the brackets of (3) in Proposition \ref{0,1,2}.

The last piece of the formula we need is a sign function $s_k:S_k\to \ZZ$. We also do this recursively. The base is $k=1$ and we set $s_1(\{1\} )=0$. For $k>1$ again we split this up differently for $I\in A_k$ and $I\in \bigcup\limits_{s=3}^{2k-1}T_k^s$. For $I\in A_k$ set $s_k(I)=|I|-1$. For $I\in T_k^s(i)$ there exists $\alpha \in A_{k-s+i}$ and $\beta \in S_i$ such that $I=\ga \sqcup \{r-s\} \sqcup F_s(\gb)$ where $s_i(\beta)$ is already defined in the context of $S_i$. Then set $s_k(I)=|\alpha |+s_i(\beta)$. This makes sense because of Lemma \ref{injectiverec}. For example, $s_2(\{2\})=0$, $s_2(\{1,2\})=1$. For $\{r-3,r-2\}$ we get that $\alpha=\emptyset$ and $\beta=\{1\}$ as seen in \ref{k=2recursion}. Hence $s_2(\{r-3,r-2\})=0+0$ and these signs again match the ``top heavy'' decomposition of the Kazhdan-Lusztig coefficient of the $k=2$ quadratic term in (3) of Proposition \ref{0,1,2}. These are all the ingredients we need for the main formula. 

\section{Top-heavy formula for Kazhdan-Lusztig polynomials}\label{thm}

In this section we state and prove the main formula which is the following theorem.

\begin{theorem}\label{KLpolymain}

For any finite, ranked lattice $\calP$ with rank $r$ the degree $k$ coefficient of the Kazhdan-Lusztig polynomial of $\calP$ for $1\leq k<r/2 $ is $$\sum\limits_{I\in S_k}(-1)^{s_k(I)}\Big( W_{t(I)}(\calP )-W_I(\calP)\Big). $$ 

\end{theorem}

\begin{proof}

Induct on $k$. The base $k=1$ is done in Proposition \ref{0,1,2}. Now we compute the degree $k$ term where $k>1$. In (3) of the recursion in Definition \ref{KLdef} the left hand side has the degree $k$ coefficient on the $t^{r-k}$ term. This is the terms that we will examine on the right hand side. First we split the right hand side up in terms of rank so that we rewrite it as \begin{equation}\label{startsum}\sum\limits_{s= 0}^r\sum\limits_{F\in \calP_{r-s}}\chi(\calP_F,t)P(\calP^F,t).\end{equation} Now we reduce this further. Suppose that $s>2k-1$. Then for $F\in \calP_{r-s}$ $\deg(\chi(\calP_F,t))=r-s$ and $\deg(P(\calP^F,t))<s/2$. So, $deg(\chi(\calP_F,t))+\deg(P(\calP^F,t))<r-s+s/2=r-s/2\leq r-k$. Hence we can reduce (\ref{startsum}) to \begin{equation}\label{2sum}\sum\limits_{s= 0}^{2k-1}\sum\limits_{F\in \calP_{r-s}}\chi(\calP_F,t)P(\calP^F,t) .\end{equation} Note that since $k<r/2$ and $\deg (P(\calP^F,t))=s$ we know that the coefficients $P(\calP^F,t)$ will all be computed by induction. For any polynomial $p$ let $u(i,p)$ denote the coefficient of the $i^{th}$ term and $d(i,p)$ be the $i^{th}$ term down from the top term (i.e. if $\deg p=d$ then $d(i,p)=u(d-i,p)$). Then for each term in (\ref{2sum}) the possible products which will yield a degree $r-k$ term are of the form $$d(k-s+i,\chi(\calP_F,t))u(i,P(\calP^F,t)) $$ where $\max \{0,s-k\}\leq i<s/2$. Hence the total coefficient we are seeking is \begin{equation}\label{totcoeff} \sum\limits_{s=0}^{2k-1}\sum\limits_{F\in \calP_{r-s}}\sum\limits_{\substack{\max\{0,s-k\}\\ \leq i<s/2}}d(k-s+i,\chi(\calP_F,t))u(i,P(\calP^F,t)) .\end{equation} 

We first focus on the terms where $i=0$. For these terms we have by Proposition \ref{0,1,2} $$ u(0,P(\calP^F,t)) =1$$ and \begin{equation}\label{1stkind}d(k-s,\chi(\calP_F,t))=w_{0,k-s}(\calP_F).\end{equation} Then we use Theorem \ref{interpolation} on (\ref{1stkind}) to get \begin{equation}\label{1stkind2}d(k-s,\chi(\calP_F,t))=\sum\limits_{I\subseteq \{1,\dots ,k-s-1\}}(-1)^{|I|+1}W_{I\cup \{k-s\}} (\calP_F).\end{equation} Notice that $0\leq s\leq k$. Summing over $F\in \calP_{r-s}$ and applying Lemma \ref{add1} to (\ref{1stkind2}) we get \begin{equation}\label{1stkind3} \sum_{F\in \calP_{r-s}}d(k-s,\chi(\calP_F,t))=\sum\limits_{I\subseteq \{1,\dots ,k-s-1\}}(-1)^{|I|+1}W_{I\cup \{k-s\}\cup \{r-s\}} (\calP) \end{equation} as long as $s\neq 0$. In that case since $\calP$ is a lattice the sum only contains one term where $\calP_F=\calP$. Hence \begin{equation}\label{nor's} d(k,\chi(\calP,t))=\sum\limits_{I\subseteq \{1,\dots ,k-1\}}(-1)^{|I|+1}W_{I\cup \{k\}}(\calP). \end{equation} The subscripts of (\ref{nor's}) give exactly all the terms of $A_k$ as well as the signs $s_k(I)$ for $I\in A_k$. Finally for each $I\subseteq A_k$ with $I=J\cup\{k-s\}\cup\{k\}$  for $1\leq s<k$ there is exactly one term in (\ref{1stkind3}), that being $J\cup\{k-s\}\cup\{r-s\}$ which is the top heavy pair to $I$. Also note that this exactly covers all the terms of (\ref{1stkind3}) and (\ref{nor's}). Hence we have verified the formula for $i=0$ and equivalently $A_k$.

Next we focus on the case where $1\leq i$. Since $i<s/2\leq k$ we have by induction that \begin{equation}\label{inducstep}u(i,P(\calP^F,t))=\sum\limits_{I\in S_i}(-1)^{s_i(I)}\Big( W_{t(I)}(\calP^F )-W_I(\calP^F)\Big).\end{equation} Because $F\in \calP_{r-s}$ we know $\rk(\calP^F)=s$ and so when using this induction all the subscripts in the formula have $r$ replaced with $s$. The terms coming from the characteristic polynomial are \begin{equation}\label{chi1}d(k-s+i,\chi(\calP_F,t))=w_{0,k-s+i}(\calP_F). \end{equation} Again using Theorem \ref{interpolation} (\ref{chi1}) becomes \begin{equation}\label{chi2}d(k-s+i,\chi(\calP_F,t))=\sum\limits_{\ga\in A_{k-s+i}}(-1)^{|\ga|}W_{\ga} (\calP_F) .\end{equation} Next putting (\ref{inducstep}) and (\ref{chi2}) together for each $i>0$ term of (\ref{totcoeff}) we get \begin{equation}\label{i>0} \sum\limits_{s=0}^{2k-1}\Bigg[ \sum\limits_{\ga \in A_{k-s+i}}(-1)^{|\ga |}W_{\ga} (\calP_F)\Bigg] \Bigg[ \sum\limits_{\gb\in S_i}(-1)^{s_i(\gb)}\Big( W_{t(\gb)}(\calP^F )-W_\gb(\calP^F)\Big)\Bigg] .\end{equation} Moving sums together (\ref{i>0}) becomes \begin{equation}\label{nexti>0} \sum\limits_{s=0}^{2k-1} \sum\limits_{\ga\in A_{k-s+i}}\sum\limits_{\gb\in S_i}(-1)^{|\ga|+s_i(\gb)}\Big(W_{\ga} (\calP_F) W_{t(\gb)}(\calP^F )-W_{\ga} (\calP_F)W_\gb(\calP^F)\Big) .\end{equation} Then summing over $F\in \calP_{r-s}$ and applying Lemma \ref{whitprod} to (\ref{nexti>0}) we get \begin{equation}\label{i>0+2}  \sum\limits_{s=0}^{2k-1} \sum\limits_{\ga\in A_{k-s+i}}\sum\limits_{\gb\in S_i}(-1)^{|\ga|+s_i(\gb)}\Big(W_{\ga\cup \{r-s\} \cup t(\gb)[r-s]}(\calP )-W_{\ga\cup \{r-s\} \cup \gb [r-s]} (\calP )\Big) .\end{equation} This all makes sense because inside $\calP^F$ the rank is $s$ and all the elements of every $\gb$ above are $< s$. Hence in the total lattice $\calP$ we can add $r-s$ and satisfy the hypothesis of Lemma \ref{whitprod}. Finally to finish the proof note that $t(\gb)[r-s]=t(F_s(\gb))$ and $\gb [r-s]=F_s(\gb)$. \end{proof} 

In Table \ref{k=1..5} we print the formula from Theorem \ref{KLpolymain} for $k=\{1,2,3,4,5\}$. For $k=6$ the formula takes up too much space. This was calculated using Sage (see \cite{sage}). There we only list the index set $S(k)$ and the corresponding sign $s_k(I)$.

{\small \begin{longtable}{|l|l|}

\hline

k & $s_k(I) I $ for $I$ in $S(k)$ \\

\hline\hline

1& +[1]\\

\hline

2& +[2], +[r - 3, r - 2], -[1, 2]\\

\hline

3& +[ 3],
 -[ 1, 3],
 -[ 2, 3],
 +[ 1, 2, 3],
 -[ 1, r - 3, r - 2],
 +[ r - 4, r - 3],
 +[ r - 5, r - 3],\\
 &
 +[ r - 5, r - 3, r - 2],
 -[ r - 5, r - 4, r - 3]\\
 
\hline

4 & +[ 4],
 -[ 1, 4],
 -[ 2, 4],
 +[ 1, 2, 4],
 -[ 3, 4],
 +[ 1, 3, 4],
 +[ 2, 3, 4],
 -[ 1, 2, 3, 4],\\
 &

 -[ 2, r - 3, r - 2],
 +[ 1, 2, r - 3, r - 2],
 -[ 1, r - 4, r - 3],
 +[ r - 5, r - 4],
-[1, r - 5, r - 3],\\
&

 -[1, r - 5, r - 3, r - 2],
 +[1, r - 5, r - 4, r - 3],
 +[ r - 6, r - 4],\\
 &

 +[ r - 6, r - 3, r - 2],
 -[ r - 6, r - 5, r - 4],
 +[ r - 7, r - 4],
 -[ r - 7, r - 6, r - 4],\\
 &

 -[ r - 7, r - 5, r - 4],
 +[ r - 7, r - 6, r - 5, r - 4],
 -[ r - 7, r - 6, r - 3, r - 2],\\
 &

 +[ r - 7, r - 4, r - 3],
 +[ r - 7, r - 5, r - 3],
 +[ r - 7, r - 5, r - 3, r - 2],\\
 &

 -[ r - 7, r - 5, r - 4, r - 3]\\
 
 \hline
 
 5&
 +[ 5],
 -[ 1, 5],
 -[ 2, 5],
 +[ 1, 2, 5],
 -[ 3, 5],
 +[ 1, 3, 5],
 +[ 2, 3, 5],
 -[ 1, 2, 3, 5],\\
 &

 -[ 4, 5],
 +[ 1, 4, 5],
 +[ 2, 4, 5],
 -[ 1, 2, 4, 5],
 +[ 3, 4, 5],
 -[ 1, 3, 4, 5],
 -[ 2, 3, 4, 5],\\
 &

 +[ 1, 2, 3, 4, 5],
 -[ 3, r - 3, r - 2],
 +[ 1, 3, r - 3, r - 2],
 +[ 2, 3, r - 3, r - 2],\\
 &

 -[ 1, 2, 3, r - 3, r - 2],
 -[ 2, r - 4, r - 3],
 +[ 1, 2, r - 4, r - 3],
 -[ 1, r - 5, r - 4],\\
 &

 +[ r - 6, r - 5],
 +[ r - 7, r - 5],
 +[ r - 7, r - 3, r - 2],
 -[ r - 7, r - 6, r - 5],\\
 &

 +[ r - 8, r - 5],
 -[ r - 8, r - 7, r - 5],
 -[ r - 8, r - 6, r - 5],
 +[ r - 8, r - 7, r - 6, r - 5],\\
 &

 -[ r - 8, r - 7, r - 3, r - 2],
 +[ r - 8, r - 4, r - 3],
 +[ r - 8, r - 5, r - 3],\\
 &

 +[ r - 8, r - 5, r - 3, r - 2],
 -[ r - 8, r - 5, r - 4, r - 3],
 +[ r - 9, r - 5],\\

&

 -[ r - 9, r - 8, r - 5],
 -[ r - 9, r - 7, r - 5],
 +[ r - 9, r - 8, r - 7, r - 5],\\
 &

 -[ r - 9, r - 6, r - 5],
 +[ r - 9, r - 8, r - 6, r - 5],
 +[ r - 9, r - 7, r - 6, r - 5],\\
 &

 -[ r - 9, r - 8, r - 7, r - 6, r - 5],
 -[ r - 9, r - 7, r - 3, r - 2],\\
 &

 +[ r - 9, r - 8, r - 7, r - 3, r - 2],
 -[ r - 9, r - 8, r - 4, r - 3],\\
 &

 +[ r - 9, r - 5, r - 4],
 +[ r - 9, r - 6, r - 4],
 +[ r - 9, r - 6, r - 3, r - 2],\\
 &

 -[ r - 9, r - 6, r - 5, r - 4],
 +[ r - 9, r - 7, r - 4],
 -[ r - 9, r - 7, r - 6, r - 4],\\
 &

 -[ r - 9, r - 7, r - 5, r - 4],
 +[ r - 9, r - 7, r - 6, r - 5, r - 4],\\
 &

 -[ r - 9, r - 7, r - 6, r - 3, r - 2],
 +[ r - 9, r - 7, r - 4, r - 3],\\
 &

 +[ r - 9, r - 7, r - 5, r - 3],
 +[ r - 9, r - 7, r - 5, r - 3, r - 2],\\
 &

 -[ r - 9, r - 7, r - 5, r - 4, r - 3] \\
 
 \hline

\caption{Low degree coefficient formulas for the matroid KL polynomial}\label{KLpolytable}\label{k=1..5}
\end{longtable}}

\begin{remark}

Note that Lemma \ref{injectiverec} shows that there is exactly one term of each type in the formula of Theorem \ref{KLpolymain}.

\end{remark}

The top heavy pairs decomposition of Theorem \ref{KLpolymain} together with the ``Hyperplane Theorem'' (\cite{A87}, Proposition 8.5.1) gives an easy proof of the ``hard'' implication of Proposition 2.14  of \cite{EPW15}.

\begin{corollary}

If $L$ is a modular lattice then $P(L,t)=1$.

\end{corollary}

\begin{remark}

Recently in \cite{HW17} June Huh and Botong Wang proved that if the poset is the lattice of flats of a realizable matroid then each term in Theorem \ref{KLpolymain} is positive (this was called the ``top heaviness conjecture'', see \cite{Kung86}). Also, it is conjectured that each of the coefficients themselves are positive for matroids (see \cite{EPW15}). However many of the signs $s_k$ are negative and at the moment we do not see a general relationship between these conjectures other than the formula of Theorem \ref{KLpolymain} itself. A nice line of future research would be to investigate how Theorem \ref{KLpolymain} and Huh and Wang's result could prove the non-negativity of the coefficients of the matroid Kazhdan-Lusztig polynomials.

\end{remark}

\begin{remark}

The function $\alpha:[n]\to \Z$ defined by $\alpha(I)=W_I$ is called the \emph{flag f-vector} in the literature (see \cite{Karu06} and \cite{Stan94}). The flag f-vector is used in the original definition of the so called \emph{cd-index} of the poset $\calP$. One possible problem for future research could be to find a relationship between the cd-index and the matroid Kazhdan-Lusztig polynomial using Theorem \ref{KLpolymain}.

\end{remark}

\noindent {\bf Acknowledgments:}

The author would like to give a special thanks to Nick Proudfoot for many helpful discussions and suggesting the existence of the formula in Theorem \ref{KLpolymain} which motived all this work. The author is grateful to the University of Oregon, Centro di Ricerca Matematica Ennio De Giorgi, and Hokkaido University for their hospitality during which some of this work was conducted. For helpful contributions the author would also like to thank Graham Denham, Ben Young, Ben Elias, Takuro Abe, Hiroaki Terao, Masahiko Yoshinaga, Joseph Kung, Thomas Zaslavsky, Laura Anderson, Victor Reiner, June Huh, Francesco Brenti, and Emanuele Delucchi. The author also wants to thank the anonymous referees for helpful suggestions and corrections.







%\bibliographystyle{plain}
%    Insert the bibliography data here.

%\bibliography{whitneyb}




\end{document}  
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

